Author's response to reviews

Title: Transient traumatic isolated neurogenic ptosis after a mild head trauma: a case report and literature review

Authors:

Guichen Li (Gretchen666@yeah.net)
Yang Zhang (zhangyang_930@126.com)
Xiaobo Zhu (zhuxiaobo@126.com)
Kun Hou (hkyyayz@yeah.net)

Version: 5
Date: 27 October 2015

Author's response to reviews: see over
Dear editor,

We want to thank you for your thoughtful and often enthusiastic comments regarding our submitted manuscript. We appreciate the opportunity to respond. Below, we respond to each comment by the editor. In addition, changes and additions to the manuscript are highlighted in red. We look forward to hearing from you soon.

Yours sincerely,

Kun Hou M.S.

Corresponding author

E-mail: hkyyayz@yeah.net
*Revision Note*

1. **Question:** "In the author contributions section we would ask you to clarify that all the contributors currently listed meet the ICMJE guidelines for authorship. Please see the following for further details:

http://www.biomedcentral.com/bmcophthalmol/authors/instructions/case

In addition we note that contributions include statistical analysis, but there is no statistical analysis in the manuscript as this is a case report. Please revise this section in light of the author guidelines. Contributors who do not meet the criteria for authorship may be thanked in the acknowledgement section. As a minor point please use author initials rather than full names in this section."

**Response:** Thanks for your advice, we have revised the authors’ contribution section according to the ICMJE guidelines for authorship and the actual contributions the authors made.

2. **Question:** Please include a list of abbreviations that were used on the manuscript. This should follow the conclusion of the main article section.

**Response:** We have added a list of abbreviations behind the conclusion section.